Author’s response to reviews

Title: Rationale and Design of a Statewide Cohort to Examine Efficient Resource Utilization for Patients with Intracerebral Hemorrhage (EnRICH)

Authors:

Farhaan Vahidy (Farhaan.Vahidy@uth.tmc.edu)
Elizabeth Meyer (elizabeth.g.meyer@uth.tmc.edu)
Arvind Bambhroliya (Arvind.B.Bambhroliya@uth.tmc.edu)
Jennifer Meeks (jennifer.r.meeks@uth.tmc.edu)
Charles Begley (Charles.E.Begley@uth.tmc.edu)
Tzu-Ching Wu (Tzu-Ching.Wu@uth.tmc.edu)
Jon Tyson (Jon.E.Tyson@uth.tmc.edu)
Charles Miller (Charles.C.Miller@uth.tmc.edu)
Ritvij Bowry (Ritvij.Bowry@uth.tmc.edu)
Wamda Ahmed (Wamda.O.Ahmed@uth.tmc.edu)
Gretchel Gealogo (GAGealogo@seton.org)
Louise McCullough (Louise.D.McCullough@uth.tmc.edu)
Steven Warach (SWarach@seton.org)
Sean Savitz (Sean.I.Savitz@uth.tmc.edu)

Version: 2 Date: 09 Mar 2018

Author’s response to reviews:

A non red lined version is being uploaded. Thank you